Author's response to reviews

Title: Increased expression of vascular endothelin type B and angiotensin type 1 receptors in patients with ischemic heart disease

Authors:

Ivan Dimitrijevic (ivan.dimitrijevic@med.lu.se)
Marie-Louise Edvinsson Edvinsson (MarieLouise.L.Edvinsson@skane.se)
Qingwen Chen (cherqw@hotmail.com)
Malin Malmsjö (malin.malmsjo@med.lu.se)
Per-Ola Kimblad (Per-Ola.Kimblad@skane.se)
Lars Edvinsson (lars.edvinsson@med.lu.se)

Version: 6 Date: 22 July 2009

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team.

Formatting changes requested:

Major revisions

Reference 40 - please provide full details for this reference.

Reply: This is now done

Figures - please cite all figures in numerical order in the manuscript. Figures 2 and 4 are currently cited in the wrong position.

Reply: We have renumbered the Figures and they are now in the correct numerical order.

Minor revisions

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Reply: this is now done

Title page - please remove short title.

Reply: This is now done

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Reply: This is now done to the best of my capacity.

Sincerely

Ivan Dimitrijevic, MD